Dynamic Complementarities between Early- and Late-Life Exposures: Evidence from the Social Security Notch^{*}

Hamid Noghanibehambari[†]

Jason Fletcher[‡]

Abstract

An old and debated research examines the income-mortality relationship and finds mixed evidence. In this paper, we extend the literature in two ways. First, we re-evaluate the previous studies using a new dataset and implementing a Notch in social security retirement benefits to overcome selection and endogeneity issues. We find a positive impact of income on longevity among the elderly population with more pronounced effects among blacks, low educated, and people of low socioeconomic status. Second, we examine the dynamic complementarities of the income shock during old ages with several early-life and childhood exposures. We find a protective role of child labor and compulsory schooling laws in the state-of-birth and adolescence years in buffering the effects of an income shock. Among blacks, exposure to early 20th century Rosenwald school construction reverses the negative income effects. Finally, we provide evidence from the Dust Bowl and New Deal relief spending that economic shocks during childhood and early adulthood can exacerbate the negative income shocks in old age, while government social spending mitigates the negative impacts.

Keywords: Social Security, Retirement, Pension, Mortality, Longevity, Historical Data, Education, Compulsory Schooling, Dynamic Complementarities, Dust Bowl, Rosenwald **JEL Codes**: H51, H53, H55, H75, I18, J18, J26, N32

Email: noghanibeham@wisc.edu, Phone: +1-806-620-1812, ORCID ID: https://orcid.org/0000-0001-7868-2900

[‡] La Follette School of Public Affairs, University of Wisconsin-Madison, 1225 Observatory Drive, Madison, WI 53706-1211, USA

Email: jason.fletcher@wisc.edu

^{*} The authors claim that they have no conflict of interest to report. The authors would like to acknowledge financial support from NIA grants (R01AG060109, R01AG076830) and the Center for Demography of Health and Aging (CDHA) at the University of Wisconsin-Madison under NIA core grant P30 AG17266.

[†] Center for Demography of Health and Aging, University of Wisconsin-Madison, 1180 Observatory Drive, Madison, WI 53706

1. Introduction

Life expectancy in the developed world has experienced a dramatic change during the last century since 1800 by roughly 45 years. This so-called *demographic transition* was accompanied by increases in output, rises in income, sharp improvements in public health, and innovations in drugs and medical technology (Eggleston & Fuchs, 2012; Lee & Reher, 2011). Various studies that span several disciplines have explored the potential sources of improvements in longevity (Lichtenberg, 2004).

A strand of this literature assesses the association between public health improvements, sanitations, vaccines, pharmaceutical innovations, and other healthcare technology changes with mortality and longevity (Baird et al., 2016; Bleakley, 2007; Crimmins & Finch, 2006; Rappuoli, 2014). An old and intensely debated body of literature explores the role of income-health gradient and specifically the role of income in mortality outcomes (Altenderfer, 1947; Cutler et al., 2006; Hupfeld, 2011; Lathrop, 1919; Lefèbvre et al., 2018). Cross-country analyses usually find an inverse relationship between measures of income per capita and mortality outcomes (Baird et al., 2011; O'Hare et al., 2013). However, the literature provides mixed results within a country and across individuals. For instance, studies document a within-month mortality cycle where mortality risks are higher at the beginning of a month following income receipt (Evans & Moore, 2012; Phillips et al., 1999). In addition, several studies document the procyclical nature of mortality (Ariizumi & Schirle, 2012; Coile et al., 2014; Edwards, 2008; Miller et al., 2009; Neumayer, 2004; Ruhm, 2000). Studies that directly examine the impact of personal income on longevity usually find a positive and robust link, suggesting a protective effect of income on mortality (Chetty et al., 2016; Gonzalez & Quast, 2013; Kinge et al., 2019; Kitagawa & Hauser, 1973; Lindahl, 2005). These studies find that, in the case of the US, inequality in life expectancy and longevity between poor and rich individuals rises as income rises. Moreover, this gap has widened over the past several decades (Chetty et al., 2016; Cristia, 2009). In contrast, other studies find zero effects or even a positive impact of income receipt on mortality (Ahammer et al., 2017; Andersson et al., 2015; Dobkin & Puller, 2007; Engelhardt et al., 2022; Evans & Moore, 2011; Snyder & Evans, 2006).

Isolating the effects of income on health and mortality is challenging for two primary reasons. First, health and income could both be the output of other processes subjecting the gradient to bias due to omitted factors. For instance, individuals with higher discount rates are also healthier, invest more in human capital, and thus have a higher income (Fuchs, 1980; Paulden & Claxton, 2012; Takagi et al., 2016; van der Pol, 2011). Another important example of endogeneity is individuals' health status, which affects their income and leads to differential longevity. Second, individuals may observe their expected longevity and make human capital investment decisions accordingly, affecting their lifetime income (Ferreira & Pessôa, 2007; Hoque et al., 2019; Zhang et al., 2003). This heterogenous response to changes in expected life expectancy could also impact parental investment in their children's health and human capital (Fortson, 2011; Jayachandran & Lleras-Muney, 2009). In addition, the parental investment could also differ by children's initial health endowment, which affects their future income and longevity (Almond & Mazumder, 2013; Fan & Porter, 2020; Restrepo, 2016).

In order to tackle these problems, several studies implement policy changes in welfare benefit receipt and social security income receipt as an arguably exogenous source of income change (Arno et al., 2011; Emery et al., 2012; Evans & Moore, 2011; Nelson & Fritzell, 2014; Salm, 2011; Stoian & Fishback, 2010). Their findings are, however, far from conclusive. For instance, Snyder & Evans (2006) use abrupt and large changes in social security retirement benefits as the source of income shock and show that reductions in benefits reduce mortality rates of the elderly population.

On the other end, there is a growing literature that examines the effects of early-life shocks and experiences of childhood and adolescence years on old-age health, mortality, and longevity (Almond & Currie, 2011; Bailey et al., 2016; Cutler et al., 2006; Cutler et al., 2007; Fletcher, 2015; Fletcher & Noghanibehambari, 2021; Lleras-Muney, 2005; Van Den Berg et al., 2006). In the case of the US, studies have linked various experiences in the early twentieth century to later-life mortality outcomes, including exposure to the Dust Bowl (Atherwood, 2022; Cutler et al., 2007), school spending (Aaronson et al., 2021), child labor and compulsory schooling laws (Fletcher, 2015; Lleras-Muney, 2005; Mazumder, 2008), migration (Black et al., 2015; Deryugina & Molitor, 2021), New Deal social spending (Noghanibehambari & Engelman, 2022), etc. These early-life exposures could interact with the impacts of later-life determinants of health, including income, to influence old-age mortality outcomes (Almond et al., 2018; Cunha & Heckman, 2007; J. J. Heckman, 2007). However, these *dynamic complementarities* between childhood exposures and old-age income in determining longevity have gained very little empirical attention.

In this paper, we employ a newly released social security administration death records to make a bridge between the two strands of literature. We use a Notch in the social security retirement benefit payments, which resulted in a sharp and unanticipated reduction in benefits for those born after January 1, 1917, the so-called *Notch cohort*. First, we examine the direct impact of the income shock on longevity. We implement a regression discontinuity design to compare the longevity of individuals born several months after the notch to those born several months before and find reductions in longevity of about 1.5 months. The average income reduction is about \$1120, off a mean of \$27,330 annually (in 2020 dollars) (Snyder & Evans, 2006). Therefore, our

results suggest that a \$1,000 decrease in income (in 2020 dollars) is associated with 0.11 years lower longevity in an elderly population. We show that our results are robust to a wide array of specification checks.

Moreover, we analyze the dynamic complementarities of the effects by childhood and adolescence exposures. We show that being born in states with stricter child labor laws and compulsory attendance laws has an insulating impact on the income-longevity relationships. Furthermore, we find a stronger income-longevity association among those whose county-ofresidence during the 1930s was hit by top-soil erosions and the Dust Bowl. In addition, we explore the heterogeneity based on cohorts' exposure to government welfare spending under the New Deal program. We find a weaker income-longevity association among those whose county-of-residence during the 1930s experienced higher per capita relief spending and vice versa. Finally, we show that Rosenwald school construction in the early twentieth century had a protective impact on income-longevity links.

This paper contributes to the ongoing literature in two ways. First, we provide new evidence on the income-longevity relationship using new and relatively large data sources. We find effects that contradict similar studies and provide empirical evidence to explain the observed difference. Moreover, since we exploit a Notch in social security payments in our identification strategy, our results have important policy implications as policymakers constantly revise the schedule and criteria of social security retirement payments. Second, we add to the understudied literature on dynamic complementarities in longevity. While some studies explore the interaction of various exposures for education and other health measures, the literature on longevity is virtually unexplored, specifically for the dynamic complementarities of early-versus-late-life exposures (Dioikitopoulos, 2014; Johnson & Jackson, 2019; Mbiti et al., 2019). This aspect of our

study also contains important policy implications and reveals how policies may interact with each other, even with those in effect decades earlier, to affect population health and longevity.

The rest of the paper is organized as follows. Section 2 reviews the related literature. Section 3 gives a background review of the social security policy change that generated the Notch. Section 4 introduces the data sources and econometric method. Section 5 discusses the results. We depart some concluding remarks in section 7.

2. Literature Review

2.1. Income and Mortality

In this subsection, we review the literature examining the association between income and mortality, focusing on old-age mortality.

Several studies document an inverse relationship between income and mortality outcomes. For instance, Arno et al. (2011) explore the link between social security benefits for retirement and elderly mortality. They find significant reductions in mortality rates following the initial implementation of social security in 1940 and the legislation-induced rises in benefits during the 1960s-70s. Salm (2011) examines the mortality effects of two US retirement policy changes in the years 1907 and 1912 that granted old-age pensions to Union Army veterans. He finds substantial reductions in mortality rates of those who received the pension. For instance, he finds that those exposed to the 1907 reform experience 11.5 percent reductions in age-adjusted mortality, or roughly 0.82 years additional life expectancy, conditional on age 65. Using his estimation of pension payments and converting the dollar values to 2020 dollars, his analysis suggests that a \$1,000 increase in income is associated with 1.7 years of higher longevity. Emery et al. (2012) use data from Canada and examine public pension reforms and longevity. They find that a universal public pension program significantly reduces the age-adjusted mortality rate of elderly people. Muennig et al. (2016) examine the impact of tax rebates under the Earned Income Tax Credit program on longevity and find positive effects. They suggest that for each additional life year, the program requires to invest about \$7,800. Nelson & Fritzell (2014) use welfare and mortality data from 18 countries over the years 1990-2009 and show that minimum income benefit as the minimum of country-specific social programs is associated with reductions in mortality and gains in life expectancy. Barham & Rowberry (2013) examine the effects of a conditional cash transfer program in Mexico on elderly mortality. They find that the program reduces the old-age mortality rate by about 4 percent. Chetty et al. (2016) employ the universe of tax return data between the years 2001-2014 linked with social security administration death records to examine the incomelongevity relationship. They find a strong and robust association. The effects are fairly constant across different income percentiles. For instance, each additional 5 percentile income rise is associated with about 0.8 years higher longevity past age 40. Moreover, they show that the gap in life expectancy between the top and bottom percentiles has increased over time. Kinge et al. (2019) use data from Norway to document a positive income-longevity association. They also find an increasing poor-rich gap in life expectancy between 2005 and 2015.

In contrast to the findings of the discussed papers, other studies suggest zero or positive association between income and mortality. For instance, Snyder & Evans (2006) employ US Vital Statistics death records from 1979 to 1990 and examine the impact of income on mortality. They use a policy change in 1977 that resulted in a Notch in social security retirement benefit payments. They show that those who experienced a reduction in benefit payments revealed lower mortality rates. They argue that increases in old-age post-retirement labor force participation are likely channels of positive impacts of benefit reductions.

Evans & Moore (2011) examine the income-mortality relationship using US data. They employ several sources of shocks to income, including the 2001 tax relief, social security receipt, and the receipt of Permeant Fund Divided of residents of Alaska. In all cases, they find a consistently positive association. Stoian & Fishback (2010) employ city-level data for the US during the 1930s to examine the potential role of the Old-Age Assistance (OAA) program on elderly mortality. They find improvements in mortality in naïve OLS regressions. However, they find economically small and statistically insignificant effects after implementing instrumental variable strategies to solve the omitted variables bias. (Ahammer et al., 2017) employ an instrumental variable strategy to estimate the effect of wage income on mortality using data from Austria and find zero effects. Engelhardt et al. (2022) explore the impact of social security early retirement age (EEA) on retirement income and mortality. They find that EEA reduces retirement age and increases old-age poverty. The EEA also results in small reductions in age-adjusted mortality rates.

Several studies suggest that unemployment and recessions may benefit survival (Neumayer, 2004; Ruhm, 2000, 2015; Strumpf et al., 2017; Tapia Granados et al., 2017). For instance, Ruhm (2000) documents a procyclical mortality pattern in the US. He argues that economic upturns increase smoking and obesity, reduce physical activity, and result in a less healthy diet. Granados et al. (2017) use a panel of European countries and show that mortality declined during the Great Recession compared with earlier years. On the other hand, other studies find a negative income-mortality relationship, specifically in the case of older workers. For instance, Coile et al. (2014) show that older workers who experience a recession during their 50s reveal lower longevity. Gordon & Sommers (2016) employ county-level median income and mortality data to document a strong negative association between income and mortality rates.

2.2. Social Security, Retirement, and Elderly Health

Social security retirement benefits can affect a wide range of elderly lives, which may influence their health and longevity. Engelhardt (2008) explores the role of social security benefits on elderly homeownership and finds that a large portion of rises in elderly homeownership during the last decades of the twentieth century is attributable to rises in social security payments. There is narrow evidence of mortality gains of homeownership (Laaksonen et al., 2008). Goda et al. (2011) document that positive income shocks change the utilization of long-term services away from nursing homes toward paid home care services. Tsai (2015) also shows that increases in social security retirement income raise home care utilization. Engelhardt et al. (2005) employ the Notch induced by the social security reform of 1977 (as in the current study) and show that reductions in benefits lead to increases in the likelihood of living with others, altering the living arrangements of elderly people. A small strand of literature examines the impact of different living arrangements on elderly mortality and finds mixed evidence (Feng et al., 2017; Koskinen et al., 2007; Li et al., 2009).

Changes in social security laws can also influence the post-retirement labor supply. Blau & Goodstein (2010) show that changes in social security laws that resulted in the contraction of benefits can explain almost one-half of the post-1990s rise in elderly labor force participation rates. Similarly, Vere (2011) documents that reductions in social security benefits result in more hours of work among the retired population. This rise in the labor force could influence income, social ties, and social interactions, which in turn affect elderly longevity (Beller & Wagner, 2018; Holwerda et al., 2012; Lleras-Muney, 2022; Rahman, 2010; Steptoe et al., 2013).

Since the current paper's main focus is on the effect of changes in income induced by social security law change on the mortality of elderly population, it is beneficial to review previous

research on the health effects of retirement. Generally, the evidence on the health effects of retirement is mixed and inconclusive (Bamia et al., 2008; Bloemen et al., 2017; Bozio et al., 2021; Brockmann et al., 2009; Fitzpatrick & Moore, 2018; Grøtting & Lillebø, 2020; Hagen, 2018; Hernaes et al., 2013; Hult et al., 2010; Stolzenberg, 2011). For instance, Garrouste & Perdrix (2021) reviews the literature on retirement and health. They find that studies usually report improvements in health for early retirement and ambiguous effects on physical health. However, the literature does not find evidence that later retirement affects mortality. Rose (2020) estimates a regression discontinuity design to exploit pension age and find significant effects of retirement on self-reported health. Eibich (2015) uses data from Germany and finds significant improvements in subjective measures of health. He then provides evidence that reductions in work-related stress, increases in sleep duration, and more regular physical exercise are potential mechanisms. Gorry & Slavov (2021) use data from England and employ instruments using the information on eligibility and employer-sponsored pension to instrument retirement decisions. They find mixed effects of retirement on several health biomarkers. For instance, they find insignificant reductions in Diabetes and increases in fibrinogen levels, an indicator of disease risks.

2.3. Early-Life and Childhood Exposures and Later-Life Mortality

There is increasing attention in the literature to linking early-life environment and childhood exposures to later-life outcomes and specifically old-age health and longevity (Almond et al., 2018; Almond & Currie, 2011b; Caruso & Miller, 2015; Currie & Rossin-Slater, 2015; Goodman-Bacon, 2021; Hayward & Gorman, 2004; Hoynes et al., 2016; Scholte et al., 2015; Steptoe & Zaninotto, 2020). For instance, Bailey et al. (2016) examine the long-run impact of the Mothers' Pension Program, a social program for poor single mothers, on their children's life expectancy later in life. They find that children of beneficiary mothers live almost one year longer

lives. Aizer et al. (2020) explore the long-run effects of a large-scale youth training program and find that enrollees are generally taller and reveal higher longevity. Noghanibehambari & Engelman (2022) examine the in-utero and early-life exposure to the social spending under the New Deal program on old-age longevity and find positive and significant effects.

A strand of this literature examines the role of early-life economic conditions on later-life mortality. Duque & Schmitz (2021) and Schmitz & Duque (2022) explore the effect of early-life local economic conditions and later-life mortality and find significant associations. Specifically, those born in states and years with more favorable economic conditions live longer lives in old ages. Cutler et al. (2007) explore the impact of reductions in income during the Dust Bowl in the 1930s on old-age health and find insignificant associations. Atherwood (2022) focus on state-level exposure measures of the Dust Bowl of young adults and explore the effects on their old-age longevity. He finds that those born in areas that were hit harder by the Dust Bowl live slightly longer lives. In more rigorous regressions, he finds effects that are indistinguishable from zero. Noghanibehambari & Fletcher (2022) revisits this question using county-level measures of exposure and document negative and significant effects.

Several studies explore the role of educational policies on later-life health and longevity. Fletcher et al. (2021) employ a sibling comparison strategy and control for the polygenic score to explore the effects of education on old age cognition score. They find economically large and significant effects of education on old-age cognition. Aaronson et al. (2021) explore the impact of exposure to Rosenwald school construction in the era of Jim Crow laws on later-life mortality and find that exposed cohorts reveal 2-3 months higher longevity. Fletcher & Noghanibehambari (2021) explore the effect of local college expansions on education and mortality. They find that those who attend college due to the increased availability of colleges and new college openings during their adolescence have roughly 1-year higher longevity. Lleras-Muney (2005) examines the association between education and mortality. She exploits compulsory schooling laws as the source of exogenous influence on education and finds significant reductions in mortality.

3. A Short Background on Social Security Notch

Prior to the 1930s, welfare support for old age was primarily by small-scale Old Age Assistance (OAA) program implemented and administered by state and local authorities. After the Great Recession hit the US economy, the federal government intervened through the New Deal relief spending programs, including establishing a social security system in 1935. As a part of this new welfare system, Old Age and Survivors Insurance (OASI) replaced the OAA program and was designed to provide retirement benefits for the elderly. The OASI schedule depended primarily on age at retirement and the pre-retirement nominal wages.

The OASI payments remained constant afterward unless there was a new statutory change adjusting the benefits with new living costs. During the 1970s, Congress used Consumer Price Index (CPI) to index the benefit tables. In the meantime, the schedule was based on unindexed average nominal wages. This so-called double indexation resulted in substantial rises in benefits. In addition, during the 1970s, the pool of workers paying social security taxes expanded as a result of the population booms of the 1950s and 1960s, which generated surpluses in social security. Hence, Congress expanded the benefit schedule even more.

The obviously wrong double indexation coupled with high inflations of the 1970s created a horizon of insolvency for the social security system as soon as the 1980s. Congress enacted in 1977 and replaced nominal wage with an indexed wage method, resulting in reductions in benefits. The new law became effective in 1979. Congress allowed those who retired before this date to remain in the old system. Those who retired after this date are forced to be included in the new system. Hence, a Notch in benefits was created based on which cohorts born after January 1, 1917, receive substantially lower benefits. Krueger & Pischke (1992) suggest that, for a person earning average wages, the Notch results in about 13 percent reductions in benefits. Snyder & Evans (2006) find about 4 percent higher income for pre-Notch generation, or \$41 per month (in 1987 dollars).

4. Data and Econometric Method

4.1. Data Sources

Our primary data source is the Death Master Files (DMF) extracted from (Goldstein et al., 2021). The DMF data contains male death records of the social security administration linked (at the individual level) to the full-count 1940 census. The data covers death between the years 1975-2005. The advantage of this data is its link to the 1940 census, which allows us to exploit information on family characteristics, individual education and socioeconomic status, and granular county-level geographic information for the years 1935 and 9140. We employ this information to infer county-of-residence during childhood and early adulthood in our analyses of dynamic complementarities. Another advantage of this data source is that it contains millions of observations, which makes it superior to many other data sources such as the Health and Retirement Study, National Longitudinal Mortality Study, etc. The large sample size also allows for exploring heterogeneity across subsamples while still maintaining statistical power. The only sample restriction is regarding cohort cut-off. For the main analysis of the paper, we restrict the sample to cohorts born up to 10 months before and after the Notch. Specifically, we avoid including cohorts of 1919 as these cohorts are likely affected by in-utero and early-life exposures of Spanish Flue that may affect their old age health and longevity (Almond, 2006; Fletcher, 2018a, 2018b; Myrskylä et al., 2013).

Summary statistics of the final sample are reported in Table 1. The final sample consists of roughly 314,000 observations. The average age-at-death is approximately 911 and 905 months for pre-Notch and Notch cohorts, respectively. About 95 percent of observations are whites, and 5 percent are blacks.

For the analysis of dynamic interaction with state-of-birth compulsory schooling laws, we employ data on child labor laws and compulsory attendance laws from Acemoglu & Angrist (2000). We follow Acemoglu & Angrist (2000) and Mazumder (2008) to construct separate indices for compulsory attendance and child labor laws as follows. Compulsory Attendance (CA) is a measure of state-imposed mandatory years of schooling and is calculated as the largest of required years of schooling before dropping out and the difference between minimum school-leaving age and maximum age at enrollment. The child Labor (CL) index measures the enforcement of age limitation for a work permit and is the largest of years of education required for a work permit and the difference between the minimum age for a work permit and the maximum age allowed for school enrollment. We then break these two indices into several dummy variables: CA8 (CA \leq 8), CA9 (CA=9), CA10 (CA \geq 10), CL6 (CL \leq 6), CL7 (CL=7), and CL8 (CL \geq 8). We merge this data to BUNMD data based on state-of-birth and the year individuals turn 14, an age at which laws could be binding.

The county-level average New Deal spending data is extracted from Fishback & Kantor (2018). County-level indicator of Rosenwald school comes from Aaronson & Mazumder (2011).

4.2. Econometric Method

We examine the reduced-form effect of reductions in retirement income due to the Notch generated by the social security policy reform of 1977. The assumption behind our regression is that the discontinuity generated by law is orthogonal to cohort characteristics, and the longevity of cohorts born several months before and after the Notch is unlikely to trend differently except for the effect of the Notch. Moreover, the Notch was unanticipated and could not affect the behavior of the elderly pre-retirement. For instance, it is difficult for those who are in the late stages of their career to have a discernible impact on their wage trend pre-retirement. We exploit the Notch using a regression discontinuity design as follows:

$$y_{id} = \alpha_0 + \alpha_1 Notch_d + \alpha_2 BirthDate_d + \alpha_3 X_i + \varepsilon_{id}$$
(1)

Where y is age-at-death of individual i who was born in birth date (month-year) d. The variable *Notch* is a dummy that equals one if the individual is born after January 1, 1917. We include a linear trend in birth date in all regressions. In X, we include individual gender, race, and ethnicity dummies. Finally, ε is an error term.

5. Results

5.1. Early-Life Exposures

Dust Bowl. From the late 19th and early 20th centuries, agricultural production expanded t large scales in the Great Plains. Deep plowing of virgin topsoil and removal of native grassland that used to protect the soil resulted in soil vulnerability to wind and water erosions. Starting from 1930, topsoil in several states had already lost its resistibility. As a result, wind could easily pick up dust from the topsoil and create clouds of dust known as Dust Clouds. These topsoil erosions, coupled with droughts, resulted in a huge loss in agricultural income and production for affected counties that lasted for years (Hornbeck, 2012). Several studies examine its enduring impacts on those who experienced the so-called Dust Bowl during infancy, childhood, and adulthood (Arthi, 2018; Atherwood, 2022; Cutler et al., 2007; Noghanibehambari & Fletcher, 2022). In this

subsection, we explore the potential differences of the effects of the Notch by individuals' exposures to the Dust Bowl.

We use data from Hornbeck (2011) to split counties into Dust Bowl and outside of Dust Bowl counties. We then use the information on county-of-residence in 1940 and county-ofresidence in 1935 (as reported in the 1940 census) to merge with the Dust bowl counties database.⁴ We explore the differences in longevity among those residing in Dust Bowl versus those in non-Dust Bowl counties, conditional on state fixed effects and controls. The results are reported in Table 2. The full model of column 3 suggests a reduction of about 2 months in longevity.

New Deal Programs. As the greatest recession in American history, the Great Depression started in 1929, peaked in 1933, and lasted for almost a decade. Hoover administration and later Roosevelt administration engineered the first federal social programs to help the unemployed and boost the economy. They proposed various programs during different stages of the recession with different criteria and purposes under the so-called New Deal relief programs (Fishback et al., 2001, 2006, 2007, 2011, 2010). Several studies point to potential short-run and long-run benefits of social spending in general, and specifically New Deal relief spending (Aizer et al., 2016; Hoynes et al., 2016; Modrek et al., 2022; Noghanibehambari & Engelman, 2022). Therefore, one would expect a possible heterogenous income-mortality relationship by exposure to relief spending at younger ages in the presence of dynamic complementarities.

We use county-level data from Fishback & Kantor (2018) to build a cumulative per-capita county-level measure of total relief spending between 1930 and 1940. We then assign the spending

⁴ To infer the county-of-residence during the 1930s, we stat by using the county-of-residence in 1935 as reported by individual in 1940. For those who report no migration since 1935, we use county-of-residence in 1940. For those who report they have migrated but the information on 1935 county is not available, we again use county-of-residence in 1940.

data to DMF observations based on their county-of-residence during the 1930s. We then explore the differences between longevity of individuals who live in counties with top-quartile of per capita New Deal spending and bottom-three quartiles. The results are reported in Table 3. We observe that those living in higher spending counties have, on average, 1.3 months higher longevity.

Rosenwald Schools. During the Jim Crow south, schools were racially segregated, and school availability and quality were limited for black children, specifically in southern states. In 1913, Brook Washington, an educator, and Julius Rosenwald, a businessman, initiated a matching grant program to build schools for black children in the rural south. Their initiative resulted in the construction of almost 5,000 schools in the following decades. Several studies suggest that increases in school availability and rises in school resources and quality due to the so-called Rosenwald school constructions were beneficial for black children's education, income, and other later-life health outcomes (Aaronson & Mazumder, 2011; Frisvold & Golberstein, 2011, 2013; Kreisman, 2017). We use data from Aaronson et al. (2021) to calculate the county-year exposure to the Rosenwald schools when individuals are between 5-17 (years of K-12 education). We then define a dummy that equals one if the person was exposed to school construction and zero otherwise. Since these schools were located in southern states where schools were segregated, the effects are expected to have been only on blacks. Therefore, we interact a black dummy with the Rosenwald indicator to examine the longevity differences due to exposure to the school openings among black people. The results are reported in Table 4. The exposure to Rosenwald across the whole population is correlated with an insignificant change in longevity (column 1). However, exposure to Rosenwald among black people is associated with 6.7 months increases in longevity, roughly 49 percent of black versus non-black gap in longevity (column 3).

State-level Compulsory schooling laws. Several studies examine the impact of education on mortality. Although the evidence is not always conclusive, this literature generally suggests the benefits of education for health and mortality (Albouy & Lequien, 2009; Buckles et al., 2016; Fletcher et al., 2021; Fletcher, 2015; Fletcher & Frisvold, 2015; Galama et al., 2018; Heckman et al., 2013; Hong et al., 2020; Lacroix et al., 2019; Meghir et al., 2018; Savelyev, 2020; Savelyev et al., 2022). While part of the education-health gradient runs through the labor market and income effects, there are other channels through which education may impact the longevity and buffer other health shocks. For instance, education could increase health-related knowledge, lead to healthier diets, and improve the quality of peers, all of which are shown to influence health and mortality (Cohen-Cole & Fletcher, 2008; Fletcher & Marksteiner, 2017; Nanri et al., 2017; Stringhini et al., 2010). Another strand of research points to the importance of state-level child labor and compulsory attendance laws on educational attainment, income, wage growth, financial investment decisions, retirement decisions, and health (Acemoglu & Angrist, 2000; Cole et al., 2014; Eckstein & Zilcha, 1994; Le Garrec & LHuissier, 2017; Oreopoulos, 2006, 2007).

In Table 5, we explore the differences in longevity of individuals born in states with different compulsory schooling and child labor laws, conditional on region fixed effects, trends, individual, family, and state covariates. We observe a correlation of 1.9 months of additional longevity for those residing in CL8 states (column 1). We also observe significant associations between CA7 and CA8 of about 1.3 and 2.5 months, respectively (column 2). In column 3, we allow the effects of compulsory attendance and child labor laws to compete. We observe that CA8 and CL8 both suggest a correlation of about 1.8 months that is statistically significant.

5.2. Results of Social Security Notch

Balancing Tests. One concern in implementing regression 1 is that there is a systematic difference in Notch cohorts' characteristics versus pre-Notch cohorts and what we observe in estimations of parameter α_1 is a reflection of those characteristics than the Notch effect. For instance, if there are more whites born after January 1, 1917, than those born in the later months of 1916, we may observe coefficients that underestimate (overestimate) the negative (positive) links. The main reason is that white individuals have higher health and longevity for reasons that race dummies in our regression cannot capture. To explore this source of endogeneity concern, we regress observable individual characteristics on *Notch* dummy controlling for a linear trend in birth date. The results are reported in Table 6. The estimated effects suggest very small in magnitude and statistically insignificant changes in the probability of being white, black, Hispanic, and having missing information on race. The percentage change from the outcomes' mean reveals quite small changes. For the outcome of being female, the effect suggests a 25 basis-points rise, off a mean of 0.52. Although the coefficient is significant at 10 percent, the effect suggests only a 0.5 percent change from the outcome mean.

Notch-Longevity Results. We start the main results by depicting a discontinuity in the longevity of cohorts at the Notch in Figure 3. We show the results for cohorts born 10 months preand post-Notch. The top panel includes a linear fit in birth date, and the bottom panel implements a third-degree polynomial fit. The outcome is death age in years. Since the death window of BUNMD is limited to deaths between 1977-2007, age-at-death is larger for earlier-born cohorts. This fact does not confound the estimates as the inclusion of birth date trend in regressions accounts for the linear evolution of longevity across cohorts. Moreover, we observe a similar slope in reduction of longevity across cohorts born before and after Notch, suggesting their longevity does not trend differently. At the Notch, we observe a noticeable discontinuity and a sharp drop in the longevity of Notch cohorts.

We report the results of Notch-longevity in Table 7. The marginal effect of column 1 suggests a reduction in longevity of about 1.5 months. The effect is virtually unchanged when we include individual covariates (column 2) and family covariates (column 3). The results suggest a positive association between income and longevity. This finding is in line with several studies that explore the effect of income benefits, other pension reforms, and personal and family income on health and longevity (Aguila et al., 2015; Chetty et al., 2016; Golberstein, 2015; Kinge et al., 2019; Nelson & Fritzell, 2014; Salm, 2011). However, this effect is in contrast with the findings of Snyder & Evans (2006), who use Vital Statistics death records between the years 1979-1990 and show that the Notch generated mortality gains. One explanation for the observed difference in the effects is that the Notch, as Snyder & Evans (2006) suggest, results in increases in labor force participation which in several ways could bring mortality gains and offset the negative effect of the drop in income. Once these cohorts reach their 80s and 90s and fewer can continue working, the drop in retirement income becomes effective and reduces longevity. To empirically explore this potential explanation, we employ data from decennial Censuses and American Community Survey for the years 1980-2007 extracted from Ruggles et al. (2020). The limitation of this data source is that it does not report month of birth.⁵ Therefore, we can only implement a cross-cohort comparison, conditional on covariates (race/ethnicity/gender dummies). Specifically, we compare the labor market outcomes of those born in 1917 versus 1916 in censuses 1980-1990 (covering Snyder & Evans (2006) period) and 2000-2007. The results are reported in in two panels for two

⁵ However, the 1980 census and ACS for post-2005 years report quarter of birth. Since we need to compare the results across various years, we only use year of birth which is available for all data years.

outcomes: *employed* and *being active in the labor force*. On average and conditional on covariates, post-Notch cohorts are 2.7 and 2.8 percentage-points more likely to be employed and active in the labor force when they are observed in 1980-1990, respectively. However, these effects decrease to 0.1 and 0.07 percentage-points, respectively, when we look at 200-2007 data years. In addition, the coefficients become statistically insignificant. These facts confirm our prior expectation that the positive effects of Notch on labor force activity, that could explain negative income-mortality association of Snyder & Evans (2006), become insignificant for later years, when cohorts reach their 80s and 90s. Hence, we observe the negative impact of the income reduction on longevity.

Snyder & Evans (2006) estimate that the Notch resulted in a reduction in benefits of about \$1120 per year, off a mean of \$27,330 (in 2020 dollars). Therefore, our results suggest that a \$1,000 decrease in income is associated with 0.11 years lower longevity in an elderly population (in 2020 dollars). Chetty et al. (2016) use tax return data and mortality records to estimate the income-longevity relationship. They use income percentile rather than income level and explore its associated with about 0.7-0.9 years increase in life expectancy. For those at the 10th percentile of income, this means an increase in income from \$16,100 to \$23,000 (in 2020 dollars). Therefore, at the lower income levels, they estimate that a decrease of \$1,000 in annual income is associated with 0.12 years lower longevity, almost equal to our estimated effects. However, they find a linear relationship between income percentile increase and longevity increase, suggesting a concave relationship when we look at income levels. Therefore, one would observe smaller associations at the higher income levels.

To understand the magnitude of the observed effect in Appendix Table A-1, we look at similar studies that use similar outcomes and data but explore different shocks. For instance,

Halpern-Manners et al. (2020) employ a twin fixed effect strategy to examine the impact of education on mortality. They find that an additional year of schooling is associated with about 0.3 years higher longevity. Therefore, the Notch has an effect of about 0.42 fewer years of education. Fletcher & Noghanibehambari (2021) employ similar data as the current study and explore the impact of college expansions during adolescence years on college education and later-life longevity. They find a treatment-on-treated effect on those who attended college due to a college opening of about 1-year higher longevity. Therefore, the effect of Notch can offset 13 percent of the positive effect of college education on longevity.

5.3. Early Versus Late-Life Interactions

Dust Bowl. We interact a Dust Bowl county indicator with the Notch and report the results in Table 8. We observe considerably larger effects among Dust Bowl Counties' residents. The double-interaction coefficient in column 2 suggests 2.6 months lower longevity for Notch cohorts who reside in a Dust Bowl county. The Notch effect is about 1.4 months reduction for those outside Dust Bowl counties.

New Deal Spending. To explore the interaction with New Deal spending, we interact with Notch dummy a dummy that indicates the county is at the top quartile of per capita spending over the years 1930-1940. We observe a positive and significant association between top spending county and longevity. The interaction term suggests a quite small and insignificant effect of Notch for those residing in top quartile of spending county. This suggests a degree of substitution between early and late life exposures.

Rosenwald Schools. To explore the interaction effect of the Rosenwald schools with our later-life income-mortality relationship, we include a triple-interaction of the constructed Rosenwald dummy and a black dummy with the Notch dummy. We implement regressions similar

to the main results are report the findings in Table 10. In column 1, we only include doubleinteraction of Notch with a black dummy. In column 2, we interact the Notch dummy with the Rosenwald dummy. In column 3, we include the triple interactions. We do not observe a considerable drop in longevity among blacks versus non-blacks due to the Notch (column 1). Among Notch cohorts regardless of race, Rosenwald school is correlated with increases in longevity (column 2). Based on the estimates in column 3, blacks, on average, have 11.1 months lower longevity. Non-blacks who were not exposed to the Rosenwald reveal a Notch discontinuity of about 1.7 months. On average, black people of pre-Notch cohorts who were exposed to Rosenwald schools during K-12 years live 2.9 months longer. Black people of Notch cohorts who were not exposed to the Rosenwald schools live 3 months shorter lives, although the coefficient is statistically insignificant. Finally, the triple-interaction term suggests an increase of roughly 4 months among Notch cohorts who are black and exposed to the Rosenwald schools. These findings suggest that early life improvements in educational quality and school availability could break the income-mortality relationships during old ages.

Compulsory Schooling. We explore the interaction of social security reform with statelevel educational policies during individuals' adolescent years. We discuss the construction of our measures of child labor and compulsory attendance laws in section 4.1. The results are reported in Table 11. In column 1, we interact indicators of child labor law with the Notch dummy. The results suggest larger Notch effects for those whose state-of-birth had a child labor law of fewer than 7 years, about 2.1 months drop in longevity. The interaction terms imply a protective role of CL7 (child labor=7 years) and CL8 (child labor≥8) against the income shock, with the largest effect from CL8, although both coefficients are statistically insignificant. In column 2, we interact the Notch with compulsory attendance laws. Those born in states with a compulsory schooling less than 6 (versus those in Ca7 and CA8 states) are more affected by the policy change. They reveal a reduction in longevity of about 1.9 months. We observe a positive interaction term suggesting an insulating role of compulsory schooling laws. However, all the interaction terms are imprecisely estimated and limit additional comments. In column 3, we include all interaction terms of columns 1 and 2, allowing the effects to compete with each other. Individuals born in states with the least strict laws reveal an effect size of 3.4 months reduction in longevity. The interaction terms of child labor laws converge to zero. We observe an insulating role of compulsory attendance laws that are almost equal between CA7 and CA8, and both are similar to the coefficients of column 2. This line of findings of our paper adds to our understanding of the benefits of education. Previous studies generally focus on short-term and long-term effects on only one aspect of individuals' outcomes. The findings of Table 11 extend this literature and explore its role in insulating other negative exposures even at the retirement age.

6. Discussion

To be written ...

7. Conclusion

This paper revisited the old question of the role of income in mortality with new data and new perspectives. While the literature on income-mortality is large, it provides mixed evidence and inconclusive findings (Altenderfer, 1947; Barham & Rowberry, 2013; Chetty et al., 2016; Evans & Moore, 2011; Salm, 2011; Snyder & Evans, 2006). To overcome endogeneity issues, we used a change in the social security retirement benefits policy that resulted in substantially lower benefits for cohorts born after January 1, 1917. We implemented regression discontinuity design and found that post-Notch cohorts have 1.5 months lower longevity. We showed that these effects are robust to a wide array of specification and functional form checks. Moreover, we observed

considerably larger effects among blacks, low-educated people, people of lower socioeconomic status families, and females.

We explored the dynamic complementarities with a series of early-life exposures. We found discernible interaction between the income shock during old ages and early-life/childhood exposures. We showed that being born in states with stricter child labor or compulsory attendance laws has the potential to insulate individuals from later-life income shocks. Among black people, those exposed to early 20th century Rosenwald school construction reveals higher insulation to income shocks than those not exposed. We found larger impacts of the income shock on longevity among those who experienced the Dust Bowl during childhood and adolescence. These aspects of our research provided new empirical evidence for interactions between early-life exposures and later-life shocks in determining mortality outcomes.

References

- Aaronson, D., & Mazumder, B. (2011). The impact of Rosenwald schools on black achievement. Journal of Political Economy, 119(5), 821–888. https://doi.org/10.1086/662962/SUPPL FILE/2010282DATA.ZIP
- Aaronson, D., Mazumder, B., Sanders, S. G., & Taylor, E. J. (2021). Estimating the effect of school quality on mortality in the presence of migration: Evidence from the jim crow south. *Journal of Labor Economics*, 39(2), 527–558. https://doi.org/10.1086/709783/SUPPL_FILE/17462DATA.ZIP
- Acemoglu, D., & Angrist, J. (2000). How Large Are Human-Capital Externalities? Evidence from Compulsory Schooling Laws. NBER Macroeconomics Annual, 15, 9–59. https://doi.org/10.1086/654403
- Aguila, E., Kapteyn, A., & Smith, J. P. (2015). Effects of income supplementation on health of the poor elderly: The case of Mexico. *Proceedings of the National Academy of Sciences of the United States of America*, 112(1), 70–75. https://doi.org/10.1073/PNAS.1414453112/SUPPL FILE/PNAS.201414453SI.PDF
- Ahammer, A., Thomas Horvath, G., & Winter-Ebmer, R. (2017). The effect of income on mortality—new evidence for the absence of a causal link. *Journal of the Royal Statistical Society: Series A (Statistics in Society)*, 180(3), 793–816. https://doi.org/10.1111/RSSA.12240
- Aizer, A., Eli, S., Ferrie, J., & Muney, A. L. (2016). The Long-Run Impact of Cash Transfers to Poor Families. *American Economic Review*, 106(4), 935–971. https://doi.org/10.1257/AER.20140529
- Aizer, A., Eli, S., Lleras-Muney, A., & Lee, K. (2020). Do Youth Employment Programs Work? Evidence from the New Deal. https://doi.org/10.3386/W27103
- Albouy, V., & Lequien, L. (2009). Does compulsory education lower mortality? *Journal of Health Economics*, 28(1), 155–168. https://doi.org/10.1016/J.JHEALECO.2008.09.003
- Almond, D. (2006). Is the 1918 influenza pandemic over? Long-term effects of in utero influenza exposure in the post-1940 US population. *Journal of Political Economy*, 114(4), 672–712. https://doi.org/10.1086/507154
- Almond, D., & Currie, J. (2011a). Human capital development before age five. In *Handbook of Labor Economics* (Vol. 4, Issue PART B). Elsevier. https://doi.org/10.1016/S0169-7218(11)02413-0
- Almond, D., & Currie, J. (2011b). Killing Me Softly: The Fetal Origins Hypothesis. Journal of Economic Perspectives, 25(3), 153–172. https://doi.org/10.1257/JEP.25.3.153
- Almond, D., Currie, J., & Duque, V. (2018). Childhood circumstances and adult outcomes: Act II. *Journal of Economic Literature*, *56*(4), 1360–1446.
- Almond, D., & Mazumder, B. (2013). Fetal Origins and Parental Responses. Annual Review of Economics, 5, 37–56. https://doi.org/10.1146/ANNUREV-ECONOMICS-082912-110145
- Altenderfer, M. (1947). Relationship between per capita income and mortality in the cities of 100,000 or more population. *Public Health Reports*, 62(48), 1681–1691.

- Andersson, E., Lundborg, P., & Vikström, J. (2015). Income receipt and mortality Evidence from Swedish public sector employees. *Journal of Public Economics*, 131, 21–32. https://doi.org/10.1016/J.JPUBECO.2015.08.006
- Ariizumi, H., & Schirle, T. (2012). Are recessions really good for your health? Evidence from Canada. Social Science & Medicine, 74(8), 1224–1231. https://doi.org/10.1016/J.SOCSCIMED.2011.12.038
- Arno, P. S., House, J. S., Viola, D., & Schechter, C. (2011). Social security and mortality: The role of income support policies and population health in the United States. *Journal of Public Health Policy*, 32(2), 234–250. https://doi.org/10.1057/jphp.2011.2
- Arthi, V. (2018). "The dust was long in settling": Human capital and the lasting impact of the American Dust Bowl. *Journal of Economic History*, 78(1), 196–230. https://doi.org/10.1017/S0022050718000074
- Atherwood, S. (2022). Does a prolonged hardship reduce life span? Examining the longevity of young men who lived through the 1930s Great Plains drought. *Population and Environment 2022*, 1–23. https://doi.org/10.1007/S11111-022-00398-W
- Baird, S., Friedman, J., & Schady, N. (2011). Aggregate income shocks and infant mortality in the developing world. *Review of Economics and Statistics*, 93(3), 847–856. https://doi.org/10.1162/REST a 00084
- Baird, S., Hicks, J. H., Kremer, M., & Miguel, E. (2016). Worms at Work: Long-run Impacts of a Child Health Investment. *The Quarterly Journal of Economics*, 131(4), 1637–1680. https://doi.org/10.1093/QJE/QJW022
- Bamia, C., Trichopoulou, A., & Trichopoulos, D. (2008). Age at Retirement and Mortality in a General Population SampleThe Greek EPIC Study. *American Journal of Epidemiology*, 167(5), 561–569. https://doi.org/10.1093/AJE/KWM337
- Barham, T., & Rowberry, J. (2013). Living longer: The effect of the Mexican conditional cash transfer program on elderly mortality. *Journal of Development Economics*, *105*, 226–236. https://doi.org/10.1016/J.JDEVECO.2013.08.002
- Beller, J., & Wagner, A. (2018). Loneliness, social isolation, their synergistic interaction, and mortality. *Health Psychology*, 37(9), 808–813. https://doi.org/10.1037/HEA0000605
- Black, D. A., Sanders, S. G., Taylor, E. J., & Taylor, L. J. (2015). The Impact of the Great Migration on Mortality of African Americans: Evidence from the Deep South. *American Economic Review*, 105(2), 477–503. https://doi.org/10.1257/AER.20120642
- Blau, D. M., & Goodstein, R. M. (2010). Can Social Security Explain Trends in Labor Force Participation of Older Men in the United States? *Journal of Human Resources*, 45(2), 328– 363. https://doi.org/10.3368/JHR.45.2.328
- Bleakley, H. (2007). Disease and Development: Evidence from Hookworm Eradication in the American South. *The Quarterly Journal of Economics*, *122*(1), 73–117. https://doi.org/10.1162/QJEC.121.1.73
- Bloemen, H., Hochguertel, S., & Zweerink, J. (2017). The causal effect of retirement on mortality: Evidence from targeted incentives to retire early. *Health Economics*, 26(12), e204–e218. https://doi.org/10.1002/HEC.3493

- Bozio, A., Garrouste, C., & Perdrix, E. (2021). Impact of later retirement on mortality: Evidence from France. *Health Economics*, *30*(5), 1178–1199. https://doi.org/10.1002/HEC.4240
- Bozzoli, C., Deaton, A., & Quintana-Domeque, C. (2009). Adult height and childhood disease. *Demography*, 46(4), 647–669. https://doi.org/10.1353/DEM.0.0079
- Brockmann, H., Müller, R., & Helmert, U. (2009). Time to retire Time to die? A prospective cohort study of the effects of early retirement on long-term survival. *Social Science & Medicine*, *69*(2), 160–164. https://doi.org/10.1016/J.SOCSCIMED.2009.04.009
- Buckles, K., Hagemann, A., Malamud, O., Morrill, M., & Wozniak, A. (2016). The effect of college education on mortality. *Journal of Health Economics*, 50, 99–114. https://doi.org/10.1016/J.JHEALECO.2016.08.002
- Caruso, G., & Miller, S. (2015). Long run effects and intergenerational transmission of natural disasters: A case study on the 1970 Ancash Earthquake. *Journal of Development Economics*, 117, 134–150. https://doi.org/10.1016/j.jdeveco.2015.07.012
- Chetty, R., Stepner, M., Abraham, S., Lin, S., Scuderi, B., Turner, N., Bergeron, A., & Cutler, D. (2016). The Association Between Income and Life Expectancy in the United States, 2001-2014. JAMA, 315(16), 1750–1766. https://doi.org/10.1001/JAMA.2016.4226
- Cohen-Cole, E., & Fletcher, J. M. (2008). Is obesity contagious? Social networks vs. environmental factors in the obesity epidemic. *Journal of Health Economics*, 27(5), 1382– 1387. https://doi.org/10.1016/J.JHEALECO.2008.04.005
- Coile, C. C., Levine, P. B., & McKnight, R. (2014). Recessions, older workers, and longevity: How long are recessions good for your health? *American Economic Journal: Economic Policy*, 6(3), 92–119.
- Cole, S., Paulson, A., & Shastry, G. K. (2014). Smart Money? The Effect of Education on Financial Outcomes. *The Review of Financial Studies*, 27(7), 2022–2051. https://doi.org/10.1093/RFS/HHU012
- Costa, D. L. (2012). Scarring and Mortality Selection Among Civil War POWs: A Long-Term Mortality, Morbidity, and Socioeconomic Follow-Up. *Demography*, 49(4), 1185–1206. https://doi.org/10.1007/S13524-012-0125-9
- Crimmins, E. M., & Finch, C. E. (2006). Infection, inflammation, height, and longevity. *Proceedings of the National Academy of Sciences of the United States of America*, 103(2), 498–503. https://doi.org/10.1073/PNAS.0501470103/SUPPL FILE/01470FIG4.PDF
- Cristia, J. P. (2009). Rising mortality and life expectancy differentials by lifetime earnings in the United States. *Journal of Health Economics*, 28(5), 984–995. https://doi.org/10.1016/j.jhealeco.2009.06.003
- Cunha, F., & Heckman, J. (2007). The Technology of Skill Formation. *American Economic Review*, 97(2), 31–47. https://doi.org/10.1257/AER.97.2.31
- Currie, J., & Rossin-Slater, M. (2015). Early-life origins of life-cycle well-being: research and policy implications. Journal of Policy Analysis and Management : [The Journal of the Association for Public Policy Analysis and Management], 34(1), 208–242. https://doi.org/10.1002/PAM.21805

Cutler, D., Deaton, A., & Lleras-Muney, A. (2006). The Determinants of Mortality. Journal of

Economic Perspectives, 20(3), 97-120. https://doi.org/10.1257/JEP.20.3.97

- Cutler, D. M., Miller, G., & Norton, D. M. (2007). Evidence on early-life income and late-life health from America's Dust Bowl era. *Proceedings of the National Academy of Sciences*, *104*(33), 13244–13249.
- Deryugina, T., & Molitor, D. (2021). The Causal Effects of Place on Health and Longevity. *Journal of Economic Perspectives*, 35(4), 147–170. https://doi.org/10.1257/JEP.35.4.147
- Dioikitopoulos, E. V. (2014). Aging, growth and the allocation of public expenditures on health and education. *Canadian Journal of Economics/Revue Canadienne d'économique*, 47(4), 1173–1194. https://doi.org/10.1111/CAJE.12113
- Dobkin, C., & Puller, S. L. (2007). The effects of government transfers on monthly cycles in drug abuse, hospitalization and mortality. *Journal of Public Economics*, 91(11–12), 2137–2157. https://doi.org/10.1016/J.JPUBECO.2007.04.007
- Duque, V., & Schmitz, L. (2021). The Influence of Early-Life Economic Shocks on Aging Outcomes: Evidence from the U.S. Great Depression. SSRN Electronic Journal. https://doi.org/10.2139/SSRN.3985570
- Eckstein, Z., & Zilcha, I. (1994). The effects of compulsory schooling on growth, income distribution and welfare. *Journal of Public Economics*, *54*(3), 339–359. https://doi.org/10.1016/0047-2727(94)90040-X
- Edwards, R. (2008). Who is hurt by procyclical mortality? *Social Science & Medicine*, 67(12), 2051–2058. https://doi.org/10.1016/J.SOCSCIMED.2008.09.032
- Eggleston, K. N., & Fuchs, V. R. (2012). The New Demographic Transition: Most Gains in Life Expectancy Now Realized Late in Life. *Journal of Economic Perspectives*, *26*(3), 137–156. https://doi.org/10.1257/JEP.26.3.137
- Eibich, P. (2015). Understanding the effect of retirement on health: Mechanisms and heterogeneity. *Journal of Health Economics*, 43, 1–12. https://doi.org/10.1016/J.JHEALECO.2015.05.001
- Emery, J. C. H., Matheson, J. A., Fishback, P., Auld, C., Wen, J. F., Murray, J., Matteo, D., Emery, G., Paarsch, H., Mckenzie, K., Fyie, K., Balan-Cohen, A., Grootendorst, P., Pendakur, K., Bruce, C., Beaujot, R., Mcquillan, K., Robinson, C., Davies, J., ... Matheson, J. A. (2012). Should income transfers be targeted or universal? Insights from public pension influences on elderly mortality in Canada, 1921–1966. *Canadian Journal of Economics/Revue Canadienne d'économique*, 45(1), 247–269. https://doi.org/10.1111/J.1540-5982.2011.01694.X
- Engelhardt, G. V. (2008). Social security and elderly homeownership. *Journal of Urban Economics*, 63(1), 280–305. https://doi.org/10.1016/J.JUE.2007.02.005
- Engelhardt, G. V., Gruber, J., & Kumar, A. (2022). Early Social Security Claiming and Old-Age Poverty. *Journal of Human Resources*, 57(4), 1079–1106. https://doi.org/10.3368/JHR.57.4.0119-9973R1
- Engelhardt, G. V., Gruber, J., & Perry, C. D. (2005). Social Security and Elderly Living Arrangements. *Journal of Human Resources*, *XL*(2), 354–372. https://doi.org/10.3368/JHR.XL.2.354

- Etilé, F., & Milcent, C. (2006). Income-related reporting heterogeneity in self-assessed health: evidence from France. *Health Economics*, 15(9), 965–981. https://doi.org/10.1002/HEC.1164
- Evans, W. N., & Moore, T. J. (2011). The short-term mortality consequences of income receipt. *Journal of Public Economics*, 95(11–12), 1410–1424.
- Evans, W. N., & Moore, T. J. (2012). Liquidity, Economic Activity, and Mortality. *The Review* of Economics and Statistics, 94(2), 400–418. https://doi.org/10.1162/REST_A_00184
- Fan, W., & Porter, C. (2020). Reinforcement or compensation? Parental responses to children's revealed human capital levels. *Journal of Population Economics*, 33(1), 233–270. https://doi.org/10.1007/s00148-019-00752-7
- Feng, Z., Falkingham, J., Liu, X., & Vlachantoni, A. (2017). Changes in living arrangements and mortality among older people in China. SSM - Population Health, 3, 9–19. https://doi.org/10.1016/J.SSMPH.2016.11.009
- Ferreira, P. C., & Pessôa, S. de A. (2007). The effects of longevity and distortions on education and retirement. *Review of Economic Dynamics*, 10(3), 472–493. https://doi.org/10.1016/J.RED.2007.01.003
- Finkelstein, A., Gentzkow, M., & Williams, H. (2021). Place-Based Drivers of Mortality: Evidence from Migration. *American Economic Review*, 111(8), 2697–2735. https://doi.org/10.1257/AER.20190825
- Fishback, P., & Kantor, S. (2018). New Deal Studies. Ann Arbor, MI: Inter-University Consortium for Political and Social Research [Distributor], 11–18.
- Fishback, P. V., Flores-Lagunes, A., Horrace, W. C., Kantor, S., & Treber, J. (2011). The Influence of the Home Owners' Loan Corporation on Housing Markets During the 1930s. *The Review of Financial Studies*, 24(6), 1782–1813. https://doi.org/10.1093/RFS/HHQ144
- Fishback, P. V., Haines, M. R., & Kantor, S. (2007). Births, Deaths, and New Deal Relief during the Great Depression. *The Review of Economics and Statistics*, 89(1), 1–14. https://doi.org/10.1162/REST.89.1.1
- Fishback, P. V., Horrace, W. C., & Kantor, S. (2001). The Impact of New Deal Expenditures on Local Economic Activity: An Examination of Retail Sales, 1929-1939. https://doi.org/10.3386/W8108
- Fishback, P. V., Horrace, W. C., & Kantor, S. (2006). The impact of New Deal expenditures on mobility during the great depression. *Explorations in Economic History*, 43(2), 179–222. https://doi.org/10.1016/J.EEH.2005.03.002
- Fishback, P. V, Johnson, R. S., & Kantor, S. (2010). Striking at the roots of crime: The impact of welfare spending on crime during the great depression. *The Journal of Law and Economics*, 53(4), 715–740.
- Fitzpatrick, M. D., & Moore, T. J. (2018). The mortality effects of retirement: Evidence from Social Security eligibility at age 62. *Journal of Public Economics*, 157, 121–137. https://doi.org/10.1016/j.jpubeco.2017.12.001
- Fletcher, J. M. (2015). New evidence of the effects of education on health in the US: Compulsory schooling laws revisited. *Social Science & Medicine*, *127*, 101–107.

https://doi.org/10.1016/J.SOCSCIMED.2014.09.052

- Fletcher, J. M. (2018a). New evidence on the impacts of early exposure to the 1918 influenza pandemic on old-age mortality. *Biodemography and Social Biology*, *64*(2), 123–126. https://doi.org/10.1080/19485565.2018.1501267
- Fletcher, J. M. (2018b). Examining the long-term mortality effects of early health shocks. *Applied Economics Letters*, 26(11), 902–908. https://doi.org/10.1080/13504851.2018.1520960
- Fletcher, J. M., & Frisvold, D. E. (2015). Higher Education and Health Investments: Does More Schooling Affect Preventive Health Care Use? *Journal of Human Capital*, 3(2), 144–176. https://doi.org/10.1086/645090
- Fletcher, J., & Marksteiner, R. (2017). Causal Spousal Health Spillover Effects and Implications for Program Evaluation. *American Economic Journal: Economic Policy*, 9(4), 144–166. https://doi.org/10.1257/POL.20150573
- Fletcher, J., & Noghanibehambari, H. (2021). *The Effects of Education on Mortality: Evidence Using College Expansions*. https://doi.org/10.3386/W29423
- Fletcher, J., Topping, M., Zheng, F., & Lu, Q. (2021). The effects of education on cognition in older age: Evidence from genotyped Siblings. *Social Science & Medicine*, 280, 114044. https://doi.org/10.1016/J.SOCSCIMED.2021.114044
- Fortson, J. G. (2011). Mortality Risk and Human Capital Investment: The Impact of HIV/AIDS in Sub-Saharan Africa. *The Review of Economics and Statistics*, 93(1), 1–15. https://doi.org/10.1162/REST_A_00067
- Franks, P., Gold, M. R., & Fiscella, K. (2003). Sociodemographics, self-rated health, and mortality in the US. *Social Science & Medicine*, 56(12), 2505–2514. https://doi.org/10.1016/S0277-9536(02)00281-2
- Frisvold, D., & Golberstein, E. (2011). School quality and the education-health relationship: Evidence from Blacks in segregated schools. *Journal of Health Economics*, 30(6), 1232– 1245. https://doi.org/10.1016/J.JHEALECO.2011.08.003
- Frisvold, D., & Golberstein, E. (2013). The Effect of School Quality on Black-White Health Differences: Evidence From Segregated Southern Schools. *Demography*, 50(6), 1989–2012. https://doi.org/10.1007/S13524-013-0227-Z
- Fuchs, V. R. (1980). *Time Preference and Health: An Exploratory Study*. https://doi.org/10.3386/W0539
- Galama, T. J., Lleras-Muney, A., & van Kippersluis, H. (2018). *The Effect of Education on Health and Mortality: A Review of Experimental and Quasi-Experimental Evidence*. https://doi.org/10.3386/W24225
- Garrouste, C., & Perdrix, E. (2021). Is there a consensus on the health consequences of retirement? A literature review. *Journal of Economic Surveys*. https://doi.org/10.1111/JOES.12466
- Goda, G. S., Golberstein, E., & Grabowski, D. C. (2011). Income and the utilization of long-term care services: Evidence from the Social Security benefit notch. *Journal of Health Economics*, *30*(4), 719–729. https://doi.org/10.1016/J.JHEALECO.2011.04.001

- Golberstein, E. (2015). The Effects of Income on Mental Health: Evidence from the Social Security Notch. *The Journal of Mental Health Policy and Economics*, *18*(1), 27. /pmc/articles/PMC4494112/
- Goldstein, J. R., Alexander, M., Breen, C., Miranda González, A., Menares, F., Osborne, M., Snyder, M., & Yildirim, U. (2021). Censoc Project. In *CenSoc Mortality File: Version 2.0. Berkeley: University of California*. https://censoc.berkeley.edu/data/
- Gonzalez, F., & Quast, T. (2013). Non-Linearities in the Relationship between Aggregate Income and Mortality Rates. *Eastern Economic Journal 2013 41:1*, 41(1), 51–69. https://doi.org/10.1057/EEJ.2013.36
- Goodman-Bacon, A. (2021). The Long-Run Effects of Childhood Insurance Coverage: Medicaid Implementation, Adult Health, and Labor Market Outcomes. *American Economic Review*, 111(8), 2550–2593. https://doi.org/10.1257/AER.20171671
- Gordon, S. H., & Sommers, B. D. (2016). Recessions, poverty, and mortality in the United States 1993–2012. American Journal of Health Economics, 2(4), 489–510. https://doi.org/10.1162/AJHE_A_00060/ASSET/IMAGES/LARGE/AJHE.2016.00060-FIGURE02.JPEG
- Gorry, D., & Slavov, S. N. (2021). The effect of retirement on health biomarkers. *Economics & Human Biology*, 40, 100949. https://doi.org/10.1016/J.EHB.2020.100949
- Grøtting, M. W., & Lillebø, O. S. (2020). Health effects of retirement: evidence from survey and register data. *Journal of Population Economics*, *33*(2), 671–704. https://doi.org/10.1007/S00148-019-00742-9/TABLES/11
- Hagen, J. (2018). The effects of increasing the normal retirement age on health care utilization and mortality. *Journal of Population Economics*, *31*(1), 193–234. https://doi.org/10.1007/S00148-017-0664-X/FIGURES/8
- Halpern-Manners, A., Helgertz, J., Warren, J. R., & Roberts, E. (2020). The Effects of Education on Mortality: Evidence From Linked U.S. Census and Administrative Mortality Data. *Demography*, 57(4), 1513–1541. https://doi.org/10.1007/S13524-020-00892-6
- Hayward, M. D., & Gorman, B. K. (2004). The long arm of childhood: The influence of earlylife social conditions on men's mortality. *Demography 2004 41:1*, *41*(1), 87–107. https://doi.org/10.1353/DEM.2004.0005
- Heckman, J. J. (2007). The economics, technology, and neuroscience of human capability formation. *Proceedings of the National Academy of Sciences*, 104(33), 13250–13255. https://doi.org/10.1073/PNAS.0701362104
- Heckman, J., Pinto, R., & Savelyev, P. (2013). Understanding the Mechanisms through Which an Influential Early Childhood Program Boosted Adult Outcomes. *American Economic Review*, 103(6), 2052–2086. https://doi.org/10.1257/AER.103.6.2052
- Hernaes, E., Markussen, S., Piggott, J., & Vestad, O. L. (2013). Does retirement age impact mortality? *Journal of Health Economics*, 32(3), 586–598. https://doi.org/10.1016/J.JHEALECO.2013.03.001
- Holwerda, T. J., Beekman, A. T. F., Deeg, D. J. H., Stek, M. L., Van Tilburg, T. G., Visser, P. J., Schmand, B., Jonker, C., & Schoevers, R. A. (2012). Increased risk of mortality associated with social isolation in older men: only when feeling lonely? Results from the Amsterdam

Study of the Elderly (AMSTEL). *Psychological Medicine*, 42(4), 843–853. https://doi.org/10.1017/S0033291711001772

- Hong, K., Savelyev, P. A., & Tan, K. T. K. (2020). Understanding the Mechanisms Linking College Education with Longevity. *Journal of Human Capital*, 14(3), 371–400. https://doi.org/10.1086/710221
- Hoque, M. M., King, E. M., Montenegro, C. E., & Orazem, P. F. (2019). Revisiting the relationship between longevity and lifetime education: global evidence from 919 surveys. *Journal of Population Economics*, 32(2), 551–589. https://doi.org/10.1007/S00148-018-0717-9/TABLES/17
- Hornbeck, R. (2011). Replication data for: The Enduring Impact of the American Dust Bowl: Short and Long-run Adjustments to Environmental Catastrophe. Harvard Dataverse. https://doi.org/10.7910/DVN/CLRPAV
- Hornbeck, R. (2012). The enduring impact of the American Dust Bowl: Short- and long-run adjustments to environmental catastrophe. In *American Economic Review* (Vol. 102, Issue 4, pp. 1477–1507). https://doi.org/10.1257/aer.102.4.1477
- Hoynes, H., Schanzenbach, D. W., & Almond, D. (2016). Long-run impacts of childhood access to the safety net. *American Economic Review*, 106(4), 903–934. https://doi.org/10.1257/aer.20130375
- Hult, C., Stattin, M., Janlert, U., & Järvholm, B. (2010). Timing of retirement and mortality A cohort study of Swedish construction workers. *Social Science & Medicine*, *70*(10), 1480–1486. https://doi.org/10.1016/J.SOCSCIMED.2009.10.043
- Hupfeld, S. (2011). Non-monotonicity in the longevity-income relationship. *Journal of Population Economics*, 24(1), 191–211. https://doi.org/10.1007/S00148-009-0284-1/FIGURES/3
- Jayachandran, S., & Lleras-Muney, A. (2009). Life Expectancy and Human Capital Investments: Evidence from Maternal Mortality Declines *. *Quarterly Journal of Economics*, 124(1), 349–397. https://doi.org/10.1162/qjec.2009.124.1.349
- Johnson, R., & Jackson, K. (2019). Reducing Inequality through Dynamic Complementarity: Evidence from Head Start and Public School Spending. *American Economic Journal: Economic Policy*, 11(4), 310–349. https://doi.org/10.1257/POL.20180510
- Kinge, J. M., Modalsli, J. H., Øverland, S., Gjessing, H. K., Tollånes, M. C., Knudsen, A. K., Skirbekk, V., Strand, B. H., Håberg, S. E., & Vollset, S. E. (2019). Association of Household Income With Life Expectancy and Cause-Specific Mortality in Norway, 2005-2015. JAMA, 321(19), 1916–1925. https://doi.org/10.1001/JAMA.2019.4329
- Kitagawa, E. M., & Hauser, P. M. (1973). Differential mortality in the United States. In *Differential Mortality in the United States*. Harvard University Press.
- Koskinen, S., Joutsenniemi, K., Martelin, T., & Martikainen, P. (2007). Mortality differences according to living arrangements. *International Journal of Epidemiology*, 36(6), 1255– 1264. https://doi.org/10.1093/IJE/DYM212
- Kreisman, D. (2017). The Next Needed Thing. *Journal of Human Resources*, *52*(2), 573–620. https://doi.org/10.3368/JHR.52.2.0413-5571R1

- Krueger, A. B., & Pischke, J.-S. (1992). The Effect of Social Security on Labor Supply: A Cohort Analysis of the Notch Generation. *Journal of Labor Economics*, *10*(4), 412–437. https://doi.org/10.1086/298294
- Kyyrä, T., & Pesola, H. (2020). Long-term effects of extended unemployment benefits for older workers. *Labour Economics*, 62, 101777. https://doi.org/10.1016/J.LABECO.2019.101777
- Laaksonen, M., Martikainen, P., Nihtilä, E., Rahkonen, O., & Lahelma, E. (2008). Home ownership and mortality: a register-based follow-up study of 300 000 Finns. *Journal of Epidemiology & Community Health*, 62(4), 293–297. https://doi.org/10.1136/JECH.2007.061309
- Lacroix, G., Laliberté-Auger, F., Michaud, P.-C., & Parent, D. (2019). The effect of college education on health and mortality: Evidence from Canada. *Health Economics*. https://doi.org/10.1002/HEC.3975
- Lathrop, J. C. (1919). Income and infant mortality. *American Journal of Public Health*, 9(4), 270–274.
- Le Garrec, G., & LHuissier, S. (2017). Differential mortality, aging and social security: delaying the retirement age when educational spillovers matter. *Journal of Pension Economics & Finance*, *16*(3), 395–418. https://doi.org/10.1017/S1474747215000396
- Lee, R. D., & Reher, D. S. (2011). Introduction: The Landscape of Demographic Transition and Its Aftermath. *Population and Development Review*, 37(SUPPL. 1), 1–7. https://doi.org/10.1111/J.1728-4457.2011.00375.X
- Lefèbvre, M., Pestieau, P., & Ponthiere, G. (2018). FGT Old-Age Poverty Measures and the Mortality Paradox: Theory and Evidence. *Review of Income and Wealth*, 64(2), 428–458. https://doi.org/10.1111/ROIW.12277
- Li, L. W., Zhang, J., & Liang, J. (2009). Health among the oldest-old in China: Which living arrangements make a difference? *Social Science & Medicine*, 68(2), 220–227. https://doi.org/10.1016/J.SOCSCIMED.2008.10.013
- Lichtenberg, F. R. (2004). Sources of U.S. longevity increase, 1960–2001. *The Quarterly Review* of Economics and Finance, 44(3), 369–389. https://doi.org/10.1016/J.QREF.2004.05.005
- Lindahl, M. (2005). Estimating the Effect of Income on Health and Mortality Using Lottery Prizes as an Exogenous Source of Variation in Income. *Journal of Human Resources*, *XL*(1), 144–168. https://doi.org/10.3368/JHR.XL.1.144
- Lleras-Muney, A. (2005). The Relationship Between Education and Adult Mortality in the United States. *The Review of Economic Studies*, 72(1), 189–221. https://doi.org/10.1111/0034-6527.00329
- Lleras-Muney, A. (2022). Education and income gradients in longevity: The role of policy. *Canadian Journal of Economics/Revue Canadienne d'économique*, 55(1), 5–37. https://doi.org/10.1111/CAJE.12582
- Mazumder, B. (2008). Does education improve health? A reexamination of the evidence from compulsory schooling laws. *Economic Perspectives*, *32*(Q II), 2–16.
- Mbiti, I., Muralidharan, K., Romero, M., Schipper, Y., Manda, C., Rajani, R., Bandiera, O., Bharadwaj, P., Cullen, J., Dahl, G., Dinkelman, T., Edmonds, E., Hoxby, C., Figlio, D.,

Jack, K., Jackson, K., Kerwin, J., Loyalka, P., Mcintosh, C., ... Suri, T. (2019). Inputs, Incentives, and Complementarities in Education: Experimental Evidence from Tanzania. *The Quarterly Journal of Economics*, *134*(3), 1627–1673. https://doi.org/10.1093/QJE/QJZ010

- Meghir, C., Palme, M., & Simeonova, E. (2018). Education and Mortality: Evidence from a Social Experiment. *American Economic Journal: Applied Economics*, *10*(2), 234–256. https://doi.org/10.1257/APP.20150365
- Meraya, A. M., Dwibedi, N., Innes, K., Mitra, S., Tan, X., & Sambamoorthi, U. (2018). Heterogeneous Relationships between Labor Income and Health by Race/Ethnicity. *Health Services Research*, 53, 2910–2931. https://doi.org/10.1111/1475-6773.12802
- Miller, D. L., Page, M. E., Stevens, A. H., & Filipski, M. (2009). Why are recessions good for your health? *American Economic Review*, 99(2), 122–127.
- Modrek, S., Roberts, E., Warren, J. R., & Rehkopf, D. (2022). Long-Term Effects of Local-Area New Deal Work Relief in Childhood on Educational, Economic, and Health Outcomes Over the Life Course: Evidence From the Wisconsin Longitudinal Study. *Demography*, 59(4), 1489–1516. https://doi.org/10.1215/00703370-10111856
- Muennig, P. A., Mohit, B., Wu, J., Jia, H., & Rosen, Z. (2016). Cost Effectiveness of the Earned Income Tax Credit as a Health Policy Investment. *American Journal of Preventive Medicine*, 51(6), 874–881. https://doi.org/10.1016/J.AMEPRE.2016.07.001
- Myrskylä, M., Mehta, N. K., & Chang, V. W. (2013). Early life exposure to the 1918 influenza pandemic and old-age mortality by cause of death. *American Journal of Public Health*, *103*(7), e83–e90.
- Nanri, A., Mizoue, T., Shimazu, T., Ishihara, J., Takachi, R., Noda, M., Iso, H., Sasazuki, S., Sawada, N., & Tsugane, S. (2017). Dietary patterns and all-cause, cancer, and cardiovascular disease mortality in Japanese men and women: The Japan public health center-based prospective study. *PLOS ONE*, *12*(4), e0174848. https://doi.org/10.1371/JOURNAL.PONE.0174848
- Nelson, K., & Fritzell, J. (2014). Welfare states and population health: The role of minimum income benefits for mortality. *Social Science & Medicine*, *112*, 63–71. https://doi.org/10.1016/J.SOCSCIMED.2014.04.029
- Neumayer, E. (2004). Recessions lower (some) mortality rates:: evidence from Germany. *Social Science & Medicine*, *58*(6), 1037–1047. https://doi.org/10.1016/S0277-9536(03)00276-4
- Noghanibehambari, H., & Engelman, M. (2022). Social insurance programs and later-life mortality: Evidence from new deal relief spending. *Journal of Health Economics*, 86, 102690. https://doi.org/10.1016/J.JHEALECO.2022.102690
- Noghanibehambari, H., & Fletcher, J. (2022). Dust to Feed, Dust to Grey: The Effect of In-Utero Exposure to the Dust Bowl on Old-Age Longevity. https://doi.org/10.3386/W30531
- O'Hare, B., Makuta, I., Chiwaula, L., & Bar-Zeev, N. (2013). Income and child mortality in developing countries: A systematic review and meta-analysis. *Journal of the Royal Society of Medicine*, *106*(10), 408–414. https://doi.org/10.1177/0141076813489680
- Oreopoulos, P. (2006). Estimating Average and Local Average Treatment Effects of Education when Compulsory Schooling Laws Really Matter. *American Economic Review*, 96(1), 152–

175. https://doi.org/10.1257/000282806776157641

- Oreopoulos, P. (2007). Do dropouts drop out too soon? Wealth, health and happiness from compulsory schooling. *Journal of Public Economics*, *91*(11–12), 2213–2229.
- Paulden, M., & Claxton, K. (2012). Budget allocation and the revealed social rate of time preference for health. *Health Economics*, 21(5), 612–618. https://doi.org/10.1002/HEC.1730
- Phillips, D. P., Christenfeld, N., & Ryan, N. M. (1999). An increase in the number of deaths in the United States in the first week of the month—an association with substance abuse and other causes of death. *New England Journal of Medicine*, *341*(2), 93–98.
- Rahman, M. O. (2010). Family matters: The impact of kin on the mortality of the elderly in rural Bangladesh. *Population Studies*, 53(2), 227–235. https://doi.org/10.1080/00324720308080
- Rappuoli, R. (2014). Vaccines: Science, health, longevity, and wealth. *Proceedings of the National Academy of Sciences of the United States of America*, 111(34), 12282. https://doi.org/10.1073/PNAS.1413559111/ASSET/9DAB10E3-0564-4A38-BF1C-0866A324CD56/ASSETS/PNAS.1413559111.FP.PNG
- Rasulo, D., Spadea, T., Onorati, R., & Costa, G. (2012). The impact of migration in all-cause mortality: The Turin Longitudinal Study, 1971–2005. *Social Science & Medicine*, 74(6), 897–906. https://doi.org/10.1016/J.SOCSCIMED.2011.10.045
- Restrepo, B. J. (2016). Parental investment responses to a low birth weight outcome: Who compensates and who reinforces? *Journal of Population Economics*, 29(4), 969–989.
- Rose, L. (2020). Retirement and health: Evidence from England. *Journal of Health Economics*, 73, 102352. https://doi.org/10.1016/J.JHEALECO.2020.102352
- Ruhm, C. J. (2000). Are Recessions Good for Your Health? *The Quarterly Journal of Economics*, *115*(2), 617–650. https://doi.org/10.1162/003355300554872
- Ruhm, C. J. (2015). Recessions, healthy no more? Journal of Health Economics, 42, 17-28.
- Salm, M. (2011). The Effect of Pensions on Longevity: Evidence from Union Army Veterans. *The Economic Journal*, *121*(552), 595–619. https://doi.org/10.1111/J.1468-0297.2011.02427.X
- Savelyev, P. A. (2020). Conscientiousness, Extraversion, College Education, and Longevity of High-Ability Individuals. *Journal of Human Resources*, 58(1), 0918-9720R2. https://doi.org/10.3368/JHR.58.1.0918-9720R2
- Savelyev, P. A., Ward, B. C., Krueger, R. F., & McGue, M. (2022). Health endowments, schooling allocation in the family, and longevity: Evidence from US twins. *Journal of Health Economics*, *81*, 102554. https://doi.org/10.1016/J.JHEALECO.2021.102554
- Schmieder, J. F., & Von Wachter, T. (2016). The Effects of Unemployment Insurance Benefits: New Evidence and Interpretation. *Annual Review of Economics*, *8*, 547–581. https://doi.org/10.1146/ANNUREV-ECONOMICS-080614-115758
- Schmitz, L., & Duque, V. (2022). In Utero Exposure to the Great Depression is Reflected in Late-Life Epigenetic Aging Signatures. https://doi.org/10.1101/2022.05.18.22275258
- Scholte, R. S., Van Den Berg, G. J., & Lindeboom, M. (2015). Long-run effects of gestation during the Dutch Hunger Winter famine on labor market and hospitalization outcomes.

Journal of Health Economics, 39, 17–30. https://doi.org/10.1016/J.JHEALECO.2014.10.002

- Snyder, S. E., & Evans, W. N. (2006). The effect of income on mortality: Evidence from the social security Notch. *Review of Economics and Statistics*, 88(3), 482–495. https://doi.org/10.1162/rest.88.3.482
- Steptoe, A., Shankar, A., Demakakos, P., & Wardle, J. (2013). Social isolation, loneliness, and all-cause mortality in older men and women. *Proceedings of the National Academy of Sciences of the United States of America*, 110(15), 5797–5801. https://doi.org/10.1073/PNAS.1219686110/SUPPL FILE/PNAS.201219686SI.PDF
- Steptoe, A., & Zaninotto, P. (2020). Lower socioeconomic status and the acceleration of aging: An outcome-wide analysis. *Proceedings of the National Academy of Sciences of the United States of America*, 117(26), 14911–14917. https://doi.org/10.1073/PNAS.1915741117
- Stoian, A., & Fishback, P. (2010). Welfare spending and mortality rates for the elderly before the Social Security era. *Explorations in Economic History*, 47(1), 1–27. https://doi.org/10.1016/j.eeh.2009.05.005
- Stolzenberg, R. M. (2011). Do Not Go Gentle Into That Good Night: The Effect of Retirement on Subsequent Mortality of U.S. Supreme Court Justices, 1801–2006. *Demography*, 48(4), 1317–1346. https://doi.org/10.1007/S13524-011-0065-9
- Stringhini, S., Sabia, S., Shipley, M., Brunner, E., Nabi, H., Kivimaki, M., & Singh-Manoux, A. (2010). Association of Socioeconomic Position With Health Behaviors and Mortality. *JAMA*, 303(12), 1159–1166. https://doi.org/10.1001/JAMA.2010.297
- Strumpf, E. C., Charters, T. J., Harper, S., & Nandi, A. (2017). Did the Great Recession affect mortality rates in the metropolitan United States? Effects on mortality by age, gender and cause of death. *Social Science & Medicine*, 189, 11–16.
- Takagi, D., Kondo, N., Takada, M., & Hashimoto, H. (2016). Educational attainment, time preference, and health-related behaviors: A mediation analysis from the J-SHINE survey. *Social Science & Medicine*, 153, 116–122. https://doi.org/10.1016/J.SOCSCIMED.2016.01.054
- Tapia Granados, J. A., Ionides, E. L., Granados, J. A. T., & Ionides, E. L. (2017). Population health and the economy: Mortality and the Great Recession in Europe. 26(12). https://doi.org/10.1002/HEC.3495
- Tsai, Y. (2015). Social security income and the utilization of home care: Evidence from the social security notch. *Journal of Health Economics*, 43, 45–55. https://doi.org/10.1016/J.JHEALECO.2014.10.001
- Van Den Berg, G. J., Lindeboom, M., Portrait, F., Berg, G. J. Van Den, Lindeboom, M., Portrait, F., den Berg, G. J., Lindeboom, M., & Portrait, F. (2006). Economic Conditions Early in Life and Individual Mortality. *American Economic Review*, 96(1), 290–302. https://doi.org/10.1257/000282806776157740
- van der Pol, M. (2011). Health, education and time preference. *Health Economics*, 20(8), 917–929. https://doi.org/10.1002/HEC.1655
- Vere, J. P. (2011). Social Security and elderly labor supply: Evidence from the Health and Retirement Study. *Labour Economics*, 18(5), 676–686.

https://doi.org/10.1016/J.LABECO.2011.02.001

- Waldorf, B. S. (2008). Is human capital accumulation a self-propelling process? Comparing educational attainment levels of movers and stayers. *The Annals of Regional Science 2008* 43:2, 43(2), 323–344. https://doi.org/10.1007/S00168-008-0225-Z
- Xu, W., Engelman, M., Palloni, A., & Fletcher, J. (2020). Where and When: Sharpening the lens on geographic disparities in mortality. SSM - Population Health, 12, 100680. https://doi.org/10.1016/J.SSMPH.2020.100680
- Xu, W., Topping, M., & Fletcher, J. (2021). State of birth and cardiovascular disease mortality: Multilevel analyses of the National Longitudinal Mortality Study. SSM - Population Health, 15, 100875. https://doi.org/10.1016/J.SSMPH.2021.100875
- Xu, W., Wu, C., & Fletcher, J. (2020). Assessment of changes in place of death of older adults who died from dementia in the United States, 2000–2014: a time-series cross-sectional analysis. *BMC Public Health 2020 20:1, 20*(1), 1–13. https://doi.org/10.1186/S12889-020-08894-0
- Zhang, J., Zhang, J., & Lee, R. (2003). Rising longevity, education, savings, and growth. *Journal* of Development Economics, 70(1), 83–101. https://doi.org/10.1016/S0304-3878(02)00088-3

Tables

	Birth Year = 1916		Birth Year = 1917	
Variable	Mean	SD	Mean	SD
Death Age (months)	910.97366	94.88092	904.65915	95.00414
Discontinuity in Birth Date (Birth	0	0	1	0
Date>January 1, 1917)				
White	.94625	.22553	.9453	.22739
Black	.05041	.2188	.05091	.21981
Race Missing	.00334	.05766	.00379	.06143
Birth Year	1916	0	1917	0
Birth Month	7.99117	2.56411	5.0396	2.61464
Death Year	1992.0445	7.90546	1992.2719	7.91675
Father's SEI 1 st Quartile	.116	.32023	.13073	.3371
Father's SEI 2 nd Quartile	.10788	.31023	.12178	.32703
Father's SEI 3 rd Quartile	.10855	.31108	.1251	.33083
Father's SEI 4 th Quartile	.10676	.30881	.1238	.32935
Father's SEI Missing	.60855	.48808	.54905	.49759
Mother's Education < 8 Years	.38184	.48584	.42711	.49466
Mother's Education = [8,12] Years	.0929	.2903	.10716	.30932
Mother's Education > 12 Years	.02026	.1409	.02336	.15103
Mother's Education Missing	.505	.49998	.44237	.49667
Rosenwald School County	.1691	.37484	.16797	.37384
New Deal Spending 1st Quartile	.26935	.44362	.26986	.44389
New Deal Spending 2 nd Quartile	.27733	.44768	.27479	.44641
New Deal Spending 3 rd Quartile	.25092	.43354	.25276	.4346
New Deal Spending 4th Quartile	.26512	.4414	.26177	.4396
Dust Bowl County	.26935	.44362	.26986	.44389
Child Labor = 7 Years	.11914	.32396	.11874	.32349
Child Labor ≥ 8 Years	.86652	.3401	.86791	.33859
Compulsory Attendance = 7 Years	.30238	.45929	.30198	.45912
Compulsory Attendance ≥ 8 Years	.48231	.49969	.48287	.49971
Share of Homeowners	.48291	.15552	.4843	.15569
Literacy Rate	.9392	.07814	.93992	.07471
Average Occupational Income Score	23.18494	4.0628	23.22276	4.10976
Share of Married	.59153	.0348	.59491	.03419
Observations	156,	212	157,	860

Table 1 - Summary Statistics of the Final Sample

	Outcome: Age at Death (Months)			
	(1)	(2)	(3)	
Drugt David Country	-1.47102*	-1.9864**	-1.99567**	
Dust Bowl County	(.78658)	(.85271)	(.84995)	
Observations	319885	319819	319819	
R-squared	.00281	.00432	.00448	
Mean DV	907.543	907.545	907.545	
%Change	-0.204	-0.235	-0.236	
State and Birth Year FE	Yes	Yes	Yes	
Controls	No	Yes	Yes	
State Trend	No	No	Yes	

Table 2 – Dust Bowl Exposure during Early Adulthood and Old-Age Mortality

	Outcome: Age at Death (Months)			
	(1)	(2)	(3)	
4 th Quartile of Spending	1.19247***	1.27633***	1.27696***	
County	(.40328)	(.41552)	(.41574)	
Observations	315750	315750	315750	
R-squared	.00281	.00519	.00538	
Mean DV	907.419	907.419	907.419	
%Change	0.131	0.141	0.141	
State and Birth Year FE	Yes	Yes	Yes	
Controls	No	Yes	Yes	
State Trend	No	No	Yes	

Table 3 - New Deal Spending Exposure during Early Adulthood and Old-Age Mortality

	Outcome: Age at Death (Months)			
	(1)	(2)	(3)	
Rosenwald \times Black			6.7293***	
Black		-9.70235***	-13.87154***	
		(1.19337)	(1.22418)	
Rosenwald	.11297		67913	
	(1.02396)		(.99466)	
Observations	319819	319819	319819	
R-squared	.00464	.00514	.0052	
Mean DV	907.545	907.545	907.545	

Table 4 – Exposure to Rosenwald Schools in County during K-12 Education and Old-Age Mortality

Notes. Robust standard errors, clustered on state, are in parentheses. Regressions include birth-state and birth-year fixed effects and a state-specific birth-date trend. Individual controls include dummies for race and ethnicity. Family controls include father socioeconomic status dummies, mother education dummies, and missing indicators for their missing values. County controls include share of literate people, share of married people, share of homeowners, and average occupational income score. The regressions are weighted using inverse probability weighting scheme with the weights extracted from probit regressions of a dummy indicating successful DMD-census linking on observable individual and family covariates.

	Outcome: Age at Death (Months)			
	(1)	(2)	(3)	
Child Labor = 7 Vears	.73718		.68245	
Child Labor – / Years	(.73667)		(.77428)	
Ch(1) = 1 + m > 0 Versus	1.88564***		1.77883***	
Child Labor ≥ 8 Years	(.49393)		(.4389)	
Compulsory Attendance =		1.25601**	.90204	
7 Years		(.53865)	(.65503)	
Compulsory Attendance		2.4734**	1.88544*	
≥ 8 Years		(.90565)	(.97569)	
Observations	314070	314070	314070	
R-squared	.00457	.00454	.00457	
Mean DV	907.431	907.431	907.431	

Table 5 – Exposure to Compulsory Schooling and Child Labor in State-of-Birth at Age 14 and Old-Age Mortality

Notes. Robust standard errors, clustered on state, are in parentheses. Regressions include birth-region and birthyear fixed effects and a region-specific birth-date trend. Individual controls include dummies for race and ethnicity. Family controls include father socioeconomic status dummies, mother education dummies, and missing indicators for their missing values. County controls include share of literate people, share of married people, share of homeowners, and average occupational income score. The regressions are weighted using inverse probability weighting scheme with the weights extracted from probit regressions of a dummy indicating successful DMDcensus linking on observable individual and family covariates.

	Outcomes:								
	White	Black	Other Race	Father's Years of Schooling	Father's Years of Schooling Missing	Father's SEI	Father's SEI Missing	Mother's Years of Schooling	Mother's Years of Schooling Missing
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Discontinuity	00198	.00163	.00035	04025	.00331	03477	.00525	00206	.00803**
at Notch	(.0021)	(.00204)	(.00055)	(.04063)	(.00347)	(.23804)	(.00341)	(.03628)	(.00353)
Observations	319885	319885	319885	147551	319885	135347	319885	169074	319885
R-squared	.00001	0	.00001	.00009	.00485	.00002	.00481	.00017	.00531
Mean DV	0.927	0.068	0.005	6.884	0.568	27.766	0.605	6.916	0.499
%Change	-0.214	2.400	7.042	-0.585	0.584	-0.125	0.867	-0.030	1.608

Table 6 - Balancing Tests

Notes. Robust standard errors are in parentheses. Regressions include a linear trend in birth date. The regressions are weighted using inverse probability weighting scheme with the weights extracted from probit regressions of a dummy indicating successful DMD-census linking on observable individual and family covariates.

	Outcome: Age at Death (Months)			
	(1)	(2)	(3)	
Discontinuity at Natah	-1.59269**	-1.55144**	-1.54539**	
Discontinuity at Notch	(.67185)	(.67144)	(.67157)	
Observations	319885	319885	319819	
R-squared	.00279	.00418	.00435	
Mean DV	907.543	907.543	907.545	
%Change	-0.175	-0.171	-0.170	
Trend in Birth Date	Yes	Yes	Yes	
Individual Controls	No	Yes	Yes	
Family Controls	No	No	Yes	

Table 7 – The Effect of Social Security Notch on Longevity

	Outcome: Age at Death (Months)		
	(1)	(2)	
Durt Brind Crimter	-1.45692	-1.00664	
Dust Bowl County	(1.43435)	(1.52629)	
Discontinuity of Notah	-1.5144**	-1.4152**	
Discontinuity at Noten	(.62821)	(.65533)	
Discontinuity at Notch × Dust	-1.92043**	-2.61145**	
Bowl County	(.95581)	(1.30744)	
Observations	319819	319819	
R-squared	.0046	.00476	
Mean DV	907.545	907.545	
State FE	Yes	Yes	
Birth Date Trend	Yes	Yes	
Individual/Family Controls	Yes	Yes	
State-by-Date Trend	No	Yes	

Table 8 – Interaction Between Notch and Dust Bowl

	Outcome: Age at Death (Months)		
	(1)	(2)	
Ath Ownstile of Seconding Country	1.35681***	1.35778***	
4 th Quartile of Spending County	$\begin{tabular}{ c c c c c } \hline Outcome: Age an & (1) & & & & & & & & & & & & & & & & & & &$	(.43264)	
Discontinuity of Notah	-1.60655**	-1.59061**	
Discontinuity at Notch	(.60677)	(.59782)	
Discontinuity at Notch $\times 4^{\text{th}}$	14485	14553	
Quartile of Spending County	(.63747)	(.79328)	
Observations	315750	315750	
R-squared	.0052	.00539	
Mean DV	907.419	907.419	
State FE	Yes	Yes	
Birth Date Trend	Yes	Yes	
Individual/Family Controls	Yes	Yes	
State-by-Date Trend	No	Yes	

Table 9 - Interaction Between Notch and New Deal Spending

	Outcome: Age at Death (Months)			
	(1)	(2)	(3)	
Discontinuity at Notch \times			4.03683**	
Rosenwald × Black			(1.7857)	
Discontinuity at Notch \times		2.01289*	1.18462	
Rosenwald		(1.14058)	(1.3388)	
Discontinuity at Notch \times	44344		-3.0342**	
Black	(.92804)		(1.30858)	
Diastry Decenyuald			2.91384**	
Black × Rosenwald			(1.44635)	
Discontinuity at Notch	-1.38007**	-2.12995***	-1.7357**	
Discontinuity at Noten	(.69326)	(.73028)	(.74419)	
Dlaak	-9.76899***		-11.09085***	
Black	(.74013)		(1.04149)	
Rosenwald		71385	-1.43561	
Rosenward		(1.03649)	(1.16636)	
Observations	319819	319819	319819	
R-squared	.00549	.00566	.00570	
Mean DV	907.545	907.545	907.545	
State FE	Yes	Yes	Yes	
Birth Date Trend	Yes	Yes	Yes	
Individual/Family Controls	Yes	Yes	Yes	
State-by-Date Trend	Yes	Yes	Yes	

Table 10 - Interaction Between	Notch and Rosenwald School
---------------------------------------	----------------------------

	Outcome: Age at Death (Months)			
	(1)	(2)	(3)	
Discontinuity of Notah	-2.10615**	-3.43665**	-3.4347**	
Discontinuity at Notch	(.8584)	(1.16269)	(1.1669)	
Child Labor = 7 Voors	.59471		.61195	
Child Labor – / Years	(.94145)		(.95577)	
Child Labor > 9 Vacua	1.48615**		1.3928**	
Child Labor $\geq \delta$ Years	(.63196)		(.56624)	
Discontinuity at Notch ×	.24127		.11649	
Child Labor = 7 Years	(1.07319)		(1.11606)	
Discontinuity at Notch ×	.70881		.68584	
Child Labor ≥ 8 Years	(.70245)		(.83067)	
Compulsory Attendance =		.23483	06908	
7 Years		(1.17359)	(1.06275)	
Compulsory Attendance		1.47535	1.1215	
≥ 8 Years		(1.36097)	(1.44507)	
Discontinuity at Notch ×		1.78626	1.69917	
Compulsory Attendance =		(1.22575)	(1.13734)	
7 Years				
Discontinuity at Notch ×		1.76418	1.35046	
Compulsory Attendance		(1.07515)	(1.30292)	
≥ 8 Years				
Observations	314070	314070	314070	
R-squared	.00485	.00482	.00486	
Mean DV	907.431	907.431	907.431	

 Table 11 - Interaction with Compulsory Schooling and Child Labor in State-of-Birth during Adolescence

 Years

Notes. Robust standard errors, clustered on state, are in parentheses. Regressions include region fixed effects and a region-specific birth-date linear trend. Individual controls include dummies for race and ethnicity. Family controls include father socioeconomic status dummies, mother education dummies, and missing indicators for their missing values. County controls include share of literate people, share of married people, share of homeowners, and average occupational income score. The regressions are weighted using inverse probability weighting scheme with the weights extracted from probit regressions of a dummy indicating successful DMD-census linking on observable individual and family covariates.

Figures



Figure 1 – Geographic Distributions of Shocks



Figure 2 – Geographic Distribution of Compulsory Attendance and Child Labor Laws



Figure 3 - Plotting Discontinuity at the Notch for Longevity

Appendix A

			0		•	
	Outcomes:					
	Employed			In Labor Force		
	Samples: Census 1980	Samples: Census 1990	Samples: Census 2000, ACS 2001- 2007	Samples: Census 1980	Samples: Census 1990	Samples: Census 2000, ACS 2001- 2007
	(1)	(2)	(3)	(4)	(5)	(6)
Discontinuity at Notch	.03989*** (.00221)	.0117*** (.00165)	.00112 (.00157)	.04085*** (.00222)	.01188*** (.00168)	.00077 (.00163)
Observations	184641	147004	113495	184641	147004	113495
R-squared	.05444	.01379	.0071	.05818	.0137	.00619
Mean DV	0.392	0.091	0.025	0.408	0.096	0.027
%Change	10.175	12.860	4.483	10.012	12.372	2.852

Appendix Table A-1 - Exploring Changes in Labor Force Outcomes of Elderly Due to the Notch

Notes. Robust standard errors are in parentheses. Regressions include a linear trend in birth date. Individual controls include dummies for race, gender, ethnicity, and missing indicators for their missing values. *** p<0.01, ** p<0.05, * p<0.1